Author's response to reviews

Title: Measurement properties of translated versions of neck-specific questionnaires: a systematic review.

Authors:

Jasper M Schellingerhout (j.schellingerhout@erasmusmc.nl)
Martijn W Heymans (mw.heymans@vumc.nl)
Ariane P Verhagen (a.verhagen@erasmusmc.nl)
Henrica CW de Vet (hcw.devet@vumc.nl)
Bart W Koes (b.koes@erasmusmc.nl)
Caroline B Terwee (cb.terwee@vumc.nl)

Version: 2 Date: 7 March 2011

Author's response to reviews:

Dear editor,

Hereby we submit our manuscript “Measurement properties of translated versions of neck-specific questionnaires: a systematic review.” for publication in BMC Medical Research Methodology.

All authors declare that they have no conflicts of interest, including specific financial interests and relationships and affiliations relevant to the subject of the manuscript. No financial support was received.

Yours sincerely, on behalf of all authors,

Jasper M Schellingerhout
Ariane P Verhagen
Martijn W Heymans
Bart W Koes
Henrica C W de Vet
Caroline B Terwee